Author's response to reviews

Title: An 8-gene qRT-PCR-based gene expression score that has prognostic value in early breast cancer.

Authors:

Iker Sánchez Navarro (ikersana@gmail.com)
Angelo Gámez Pozo (angelogamez@gmail.com)
Álvaro Pinto (alvaro_pinto_marin@hotmail.com)
David Hardisson (dhardisson.hulp@salud.madrid.org)
Rosario Madero (madero.hulp@salud.madrid.org)
Rocío López (rlvacas@gmail.com)
Belén San José (bsjv@gmail.com)
Pilar Zamora (zamorapilar@terra.es)
Andrés Redondo (redondo12@gmail.com)
Jaime Feliu (jfeliu.hulp@salud.madrid.org)
Paloma Cejas (palomacejas@gmail.com)
Manuel González Barón (mgonzalezb.hulp@salud.madrid.org)
Juan Ángel Fresno Vara (jafresno@gmail.com)
Enrique Espinosa (eespinosa00@terra.es)

Version: 6 Date: 27 June 2010

Author's response to reviews: see over
List of changes

Major changes
In the title page, the list of authors has been re-formatted and their full affiliation is given after each name. This page also includes e-mails.
The abstract contains the required headings.
Conclusions have been added after the discussion.
Figures have been renumbered and their legends moved to the end of the manuscript.
Figure 4 and 5 were the same and former figure 3 has been included in figure 1, so there are 3 figures now.
Additional files have been renamed as such and renumbered.

Minor changes
Capital letters are used as required throughout the text.
Running title has been removed.
Presentation information has been moved to the “acknowledgement” section.